Author’s response to reviews

Title: Anticancer therapy within the last 30 days of life: results of an audit and re-audit cycle from an Australian regional cancer centre.

Authors:

Mike Nguyen (mikemenguyen@gmail.com)
Sean Ng Ying Kin (sean.nyk93@gmail.com)
Evonne Shum (evonneshum@gmail.com)
Alysson Wann (alysson.wann@gmail.com)
Babak Tamjid (babaktamjid@hotmail.com)
Arvind Sahu (arvind.sahu@gvhealth.org.au)
Javier Torres (javier.torres@gvhealth.org.au)

Version: 2 Date: 13 Jan 2020

Author’s response to reviews:

Dear editor,

Thank you for your decision and comments. The following revisions have been made.

1. ASCO abstract acknowledged in Manuscript
2. Author contributions have been revised
3. Cover letter has been removed. Response letter to reviewers comments is no longer included in submission
4. Clean manuscript is provided